Author’s response to reviews

Title: Ocular Surface Health in Shanghai University Students: A Cross-sectional Study

Authors:
Shanshan li (shanshanli_133@163.com)
Jiangnan He (hejiangnan85@126.com)
Qiuying Chen (izzie_qiu@yeah.net)
Bilian Ke (kebilian@126.com)
Zhu Jianfeng (jiazhu1974@qq.com)
Haidong Zou (zouhaidong@263.net)
Xun XU (drxuxun@sjtu.edu.cn)

Version: 1 Date: 17 Jan 2018

Author’s response to reviews:

Many thanks to the editor who gave us the following comments! And here is our reply:

1. We note that some of your participants were <18 years old. We understand that most of the participants were freshmen and sophomores, but please clarify whether any of the participants were under the age of 16. If any of the participants in your study were under the age of consent, please clarify whether consent to participate was obtained from the parents/legal guardians of these participants

Reply to comment: There was no participant under the age of 17 at the time of participating in our study. The ages were calculated automatically once the dates of birth were entered into system. This is now clarified in the “Results” section of the revised manuscript.
2. Please include a statement about consent to participate in the 'Ethics approval and consent to participate' section of your manuscript.

Reply to comment: A statement about consent to participate is now included in the 'Ethics approval and consent to participate' section of our revised manuscript.